---
displayed_sidebar: pricingReference
---

import { TypeList } from "docs-ui"

# MoneyAmountDTO

A money amount's data. A money amount represents a price.

<TypeList types={[{"name":"id","type":"`string`","description":"The ID of the money amount.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"created_at","type":"`Date`","description":"When the money\\_amount was created.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"updated_at","type":"`Date`","description":"When the money\\_amount was updated.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"deleted_at","type":"`null` \\| `Date`","description":"When the money\\_amount was deleted.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"currency_code","type":"`string`","description":"The currency code of this money amount.","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"amount","type":"[BigNumberValue](../../types/pricing.BigNumberValue/page.mdx)","description":"The price of this money amount.","optional":true,"defaultValue":"","expandable":false,"children":[{"name":"numeric","type":"`number`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"raw","type":"[BigNumberRawValue](../../types/pricing.BigNumberRawValue/page.mdx)","description":"","optional":true,"defaultValue":"","expandable":false,"children":[{"name":"value","type":"`string` \\| `number`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]},{"name":"bigNumber","type":"`BigNumber`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]}]},{"name":"min_quantity","type":"[BigNumberValue](../../types/pricing.BigNumberValue/page.mdx)","description":"The minimum quantity required to be purchased for this price to be applied.","optional":true,"defaultValue":"","expandable":false,"children":[{"name":"numeric","type":"`number`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"raw","type":"[BigNumberRawValue](../../types/pricing.BigNumberRawValue/page.mdx)","description":"","optional":true,"defaultValue":"","expandable":false,"children":[{"name":"value","type":"`string` \\| `number`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]},{"name":"bigNumber","type":"`BigNumber`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]}]},{"name":"max_quantity","type":"[BigNumberValue](../../types/pricing.BigNumberValue/page.mdx)","description":"The maximum quantity required to be purchased for this price to be applied.","optional":true,"defaultValue":"","expandable":false,"children":[{"name":"numeric","type":"`number`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"raw","type":"[BigNumberRawValue](../../types/pricing.BigNumberRawValue/page.mdx)","description":"","optional":true,"defaultValue":"","expandable":false,"children":[{"name":"value","type":"`string` \\| `number`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]},{"name":"bigNumber","type":"`BigNumber`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]}]},{"name":"rules_count","type":"`number`","description":"The number of rules that apply to this price","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"price_rules","type":"[PriceRuleDTO](../pricing.PriceRuleDTO/page.mdx)[]","description":"The price rules that apply to this price","optional":true,"defaultValue":"","expandable":false,"children":[{"name":"id","type":"`string`","description":"The ID of the price rule.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"price_set_id","type":"`string`","description":"The ID of the associated price set.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"price_set","type":"[PriceSetDTO](../pricing.PriceSetDTO/page.mdx)","description":"The associated price set.","optional":false,"defaultValue":"","expandable":true,"children":[{"name":"id","type":"`string`","description":"The ID of the price set.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"prices","type":"[MoneyAmountDTO](./page.mdx)[]","description":"The prices that belong to this price set.","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"calculated_price","type":"[CalculatedPriceSet](../pricing.CalculatedPriceSet/page.mdx)","description":"The calculated price based on the context.","optional":true,"defaultValue":"","expandable":false,"children":[]}]},{"name":"attribute","type":"`string`","description":"The attribute of the price rule","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"value","type":"`string`","description":"The value of the price rule.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"priority","type":"`number`","description":"The priority of the price rule in comparison to other applicable price rules.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"price_id","type":"`string`","description":"The ID of the associated price.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"price_list_id","type":"`string`","description":"The ID of the associated price list.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"created_at","type":"`Date`","description":"When the price\\_rule was created.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"updated_at","type":"`Date`","description":"When the price\\_rule was updated.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"deleted_at","type":"`null` \\| `Date`","description":"When the price\\_rule was deleted.","optional":false,"defaultValue":"","expandable":false,"children":[]}]}]} expandUrl="https://docs.medusajs.com/learn/fundamentals/data-models/manage-relationships#retrieve-records-of-relation" sectionTitle="MoneyAmountDTO"/>
